Author’s response to reviews

Title: A survey of knowledge, attitudes and practice of emergency contraception among university students in Cameroon

Authors:

Eugene J Kongnyuy (kongnyuy@gmail.com)
Pius Ngassa (piusngassa2004@yahoo.co.uk)
Nelson Fomulu (nelson_fomulu@yahoo.com)
Charles S Wiysonge (wiysonge@yahoo.com)
Luc Kouam (relinkouam@yahoo.frr)
Anderson S Doh (doh_anderson@yahoo.com)

Version: 7 Date: 13 July 2007

Author’s response to reviews: see over
13 July 2007

Ann Le Good
Assistant Editor
BMC-series journals

Re: 2560267421262078 - A survey of knowledge, attitudes and practice of emergency contraception among university students in Cameroon

Dear Ann,

We are happy that our manuscript has been accepted for publication in BMC Emergency Medicine pending formatting requirements.

We have put the abbreviation “ECP” after the first usage of Emergency Contraceptive Pills in the first paragraph of the Background. Thank you for the useful comments.

We have also use sentence case for all headings and sub-headings and have inserted the heading “Tables” ahead of the tables section. Thanks for the useful comments.

We hope these corrections will satisfy the publication requirements. We are looking forward to seeing the manuscript published.

Yours sincerely,

Eugene